Author's response to reviews

Title: Relative adrenal insufficiency and hemodynamic status in cardiopulmonary bypass surgery patients. A prospective cohort study.

Authors:

JOSE L IRIBARREN (joseluis.iribarren@gmail.com)
JUAN J JIMENEZ (jjimenezrivera@gmail.com)
DOMINGO HERNANDEZ (domingohernandez@gmail.com)
LISSET LORENZO (lissetlorenzo@gmail.com)
MAITANE BROUARD (maitanebrouard@gmail.com)
ANTONIO MILENA (amilena@ull.es)
MARIA L MORA (mmora@huc.canarias.org)
RAFAEL MARTINEZ (drmartinezsanz@hotmail.com)

Version: 3 Date: 25 March 2010

Author's response to reviews: see over
*In page number 4:

Lines 15 and 18, has been added the R trademark to Synacthene and Immulite.

Line 25: 15 and mmHg has been corrected.

*In page 6, we have added these sentences:

Line 12: ‘Mean arterial pressure (MAP), systemic vascular resistance index (SVRI), systolic volume index(SVI), mixed venous saturation(SvO₂) and lactic acid were similar in both groups…’

Line 22: ‘Finally, no differences between the two groups were observed in postoperative bleeding, re-exploration, mortality and length of stay in the critical care unit’.

*In page number 11: Author's contributions: ...processing blood simples; "simples”, It has been changed for ‘samples’.

*Table 1. Now we have added percentage of re-exploration caused by bleeding and Parsonnet reference.

*In table 3 we reported time as dose (last cell row), previously we reported “Norepinephrine (mcg/kg/min) hrs”, now we have taken “(mcg/kg/min)” off.